**Author's response to reviews**

**Title:** Femoral nerve compression secondary to ganglion cyst arising from hip joint: a case report

**Authors:**

Aydiner M.D. Kalaci (orthopedi@gmail.com)
Yunus M.D. Dogramaci (yunus_latif85@hotmail.com)
Teoman Toni M.D. Sevinc (sevinctt@mynet.com)
Ahmet Nedim Prof. Yanat (an_yanat@yahoo.com)

**Version:** 4 **Date:** 26 October 2008

**Author's response to reviews:** see over
Oct, 26, 2008

Dear Editor,

We have made the recommended revisions on our manuscript entitled ‘Femoral nerve compression secondary to ganglion cyst arising from the hip joint: a case report’

1-We added more bibliography and references about ganglion and hip into the discussion part of the text as recommended by the reviewer.

2- The role of colour Doppler and conventional ultrasound has been added to the discussion.

Looking forward for your answers with the hope of reflecting our manuscript to the colleague.

Respectfully yours,

Aydoğan Kalacı M.D. Assistant Professor

Corresponding author: Aydoğan Kalacı M.D. Assistant Professor. Mustafa Kemal University

Faculty of Medicine, Dept. Of Orthopaedics and Traumatology

31100 / ANTAKYA / HATAY / TURKEY

TEL: (+90)326 2148661

FAX: (+90)326 2144977   E-MAİL: orthopedi@gmail.com